Author's response to reviews

Title: Rare association between cystic fibrosis, Chiari I malformation, and hydrocephalus in an infant: case report and literature review

Authors:

Akash J Patel (akash.patel@bcm.edu)
Viraj H Raol (viraj.raol@gmail.com)
Andrew Jea (ajea@bcm.edu)

Version: 4 Date: 19 April 2011

Author's response to reviews: see over
Dear Editor,

Thank you for your thoughtful review of our article “Rare association between cystic fibrosis, Chiari I malformation, and hydrocephalus in an infant: case report and literature review” that we submitted to the Journal of Medical Case Reports.

We have made the following suggested changes to our manuscript:

- Please restructure the Abstract into the following three sections: Introduction, Case presentation, and Conclusion.
  - We have made the suggested changes.
- Please include the ethnicity, sex and age of the patient in the abstract.
  - We have included the suggested information.
- In keeping with journal style, please remove the sub-sections/sub-headings from the case presentation section.
  - We have removed subheadings in the case presentation section.
- Blank out writing within the MRI scan of Figure 1. Please edit your figure to preserve the anonymity of the patient by removing or concealing any identifiable features, including birthmarks and tattoos.
  - We have removed the letters from Figure 1.
- Please clarify whether the table contained within additional file is original to your manuscript. If not, please clarify as to whether you have permission to use this information and please acknowledge the source of this information.
  - We have clarified that our table is original to our manuscript.

We would like to thank you for your comments and hope that you find the above changes sufficient.

Sincerely,

Andrew Jea MD

Corresponding Author